Author’s response to reviews

Title: Effects of kolaviron on hepatic oxidative stress in streptozotocin induced diabetes

Authors:

Omolola R Ayepola (lolaayepola@yahoo.com)
nicole L Brooks (brooksn@cput.ac.za)
Oluwafemi O Oguntibeju (oguntibejuo@cput.ac.za)

Version: 4 Date: 29 June 2015

Author’s response to reviews: see over
Author's response to reviewer

Title: Effects of kolaviron on hepatic oxidative stress in streptozotocin induced diabetes

Manuscript number: MS: 3732491701581902

Authors: Omolola R Oyenihi, Nicole L Brooks and Oluwafemi O Oguntibeju

Author's response to reviewer:

June 28th, 2015
Editor,
BMC Complementary and Alternative Medicine

Dear Sir,

Please find attached the revised manuscript as per the comments of the reviewer. Answers to the reviewers’ comments are provided below. Please do not hesitate to contact us should you require further clarifications.

Sincerely Yours
Dr. O.R. Oyenihi
Reviewer 1

Comment 1: The concentration of DMSO has still not been mentioned

Response 1: The concentration of DMSO has been included in the manuscript.

Comment 2: The explanation for increase in SOD in the C+KV group and also the D+KV group has not been mentioned. I feel it should be revised properly

Response 2: The explanation for increase in SOD in the C+KV group and also the D+KV group has been revised.
Reviewer 2

Comment
Page no. 4 “96-well fat bottom” it should be flat bottom.
Page no. 5 “170µL” please mentioned the correct unit.
Page no. 10 “Allium cepa” should be in italics. Same for other medicinal plants.
References should be in uniform format.
Paper must be carefully reviewed to correct any other mistake.

Response: Typographical errors have been corrected and botanical names of the medicinal plants have been italicised. References are also in uniform format.